Author's response to reviews

Title: Clinicopathologic characteristics and treatment outcomes of hepatoid adenocarcinoma of stomach, a rare but unique subtype of gastric cancer

Authors:

Sun Kyung Baek (wkiki@medimail.co.kr)
Sae-Won Han (seawonhan@snu.ac.kr)
Do-Youn Oh (phdoyoun@snu.ac.kr)
Seock-Ah Im (moisa@snu.ac.kr)
Tae-You Kim (kimty@snu.ac.kr)
Yung-Jue Bang (bangyj@snu.ac.kr)

Version: 2 Date: 20 February 2011

Author's response to reviews: see over
13 Feb, 2011

Dear Editor-in-Chief of *BMC Gastroenterology*

We deeply appreciate your careful comments for our manuscript entitled, “Clinicopathologic characteristics and treatment outcomes of hepatoid adenocarcinoma of stomach, a rare but unique subtype of gastric cancer”. Your comments were really helpful for revising our manuscript. We totally agree with the comments of reviewers. We have revised our manuscript regarding to reviewers’ comments. The changes and responses to specific comments will follow at the Author Response. We hope that these changes may satisfy the editors and reviewers.

Thank you again for giving us important comments on our work. We hope you to consider this paper for publication in *BMC Gastroenterology*

Sincerely,

Do-Youn Oh, M.D., Ph.D.
Assistant Professor
Department of Internal Medicine
Seoul National University Hospital
101 Daehang-ro, Jongno-gu, Seoul 110-744, Korea
Tel: +82-2-2072-0701
Fax: +82-2-762-9662

E-mail: ohdoyoun@snu.ac.kr